Author's response to reviews

Title: Survey of information technology in Intensive Care Units in Ontario, Canada

Authors:

Stephen E Lapinsky (stephen.lapinsky@utoronto.ca)
David Holt (djholty@gmail.com)
David Hallett (hawk.hallett@utoronto.ca)
Mohamed Abdolell (m.abdolell@gmail.com)
Neill KJ Adhikari (Neill.Adhikari@sunnybrook.ca)

Version: 5 Date: 19 January 2008

Author's response to reviews: see over
The Editor  
BioMed Central  

Re: Manuscript revision - MS: 4862713551517643  

Thank you for the reviews on the manuscript “Survey of information technology use in Intensive Care Units in Ontario, Canada”. The manuscript has been reformatted as requested:

Please remove 'MS: 4862713551517643 ' from the title page. DONE  
Please give full affiliation details for affiliation 5. DONE  
Please delete the key words from the abstract. DONE  
Introduction- please rename this section Background DONE  
Materials and Methods- please rename this section Methods DONE  
Competing interests- please change 'interest' to 'interests' DONE  
Author's roles- please rename this section Authors' contributions DONE  

References - The reference list should contain all authors¿ names, if there are 30 or below. The term ¿et al.¿ should not be used. Please update reference 14. DONE  

Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures CROPPED POWERPOINT FIGURE UPLOADED  

Website addresses – Websites must be removed from the text, and cited from the main reference list only. Please present your reference in the following structure, e.g. 17. Website title [www.website.com]. DONE  

References 7, 14 are missing from the text. Please ensure all references are cited in the text in numerical order. DONE  

Sincerely,  

Stephen Lapinsky